Author’s response to reviews

Title: A State-Level Study of Opioid Use Disorder Treatment Access and Neonatal Abstinence Syndrome

Authors:

Elizabeth Wolf (elizabeth.wolf@vcuhealth.org)
Sebastian Tong (sebastian.tong@vcuhealth.org)
Roy Sabo (roy.sabo@vcuhealth.org)
Steven Woolf (steven.woolf@vcuhealth.org)
Kassie Abbinanti (kma3au@virginia.edu)
James Pecsok (jpecsok@mymail.vcu.edu)
Alex Krist (alexander.krist@vcuhealth.org)

Version: 2 Date: 11 Sep 2019

Author’s response to reviews:

We have attached a document that responds to each comment.